Author's response to reviews

Title: Compliance of Clinical Trial Registers with the World Health Organization Minimum Data Set: A Survey

Authors:

- Lorenzo Moja (moja@marionegri.it)
- Moschetti Ivan (moschetti@marionegri.it)
- Nurbhai Munira (mnurbhai@ohri.ca)
- Anna Compagnoni (anna.compagnoni@tin.it)
- Alessandro Liberati (aleslib@mailbase.it)
- Jeremy M Grimshaw (jgrimshaw@ohri.ca)
- An-Wen Chan (anwen.chan@utoronto.ca)
- Key Dickersin (kdickers@jhsph.edu)
- Karmela Krleza-Jeric (Karmela.Krleza-Jeric@cihr-irsc.gc.ca)
- David Moher (davidmoher@gmail.com)
- Ida Sim (sim@medicine.ucsf.edu)
- Jimmy Volmink (jvolmink@cormack.uct.ac.za)

Version: 2 Date: 27 November 2008

Author's response to reviews:

- Included a separate section for Conclusions
- Formatted disclosures into a statement of Competing Interests
- Included legends to additional files in the main body of the manuscript